Author's response to reviews

Title: Sequencing of NOTCH1, GATA5, TGFBR1 AND TGFBR2 genes in familial cases of Bicuspid Aortic Valve

Authors:

Ilenia Foffa (ilenia.foffa@ifc.cnr.it)
Lamia Ait Ali (aitlamia@ifc.cnr.it)
Paola Panesi (panesi@ifc.cnr.it)
Massimiliano Mariani (Mariani@ftgm.it)
Pierluigi Festa (festa@ftgm.it)
Nicoletta Botto (botto@ftgm.it)
Cecilia Vecoli (vecoli@ifc.cnr.it)
Maria Grazia Andreassi (andreassi@ifc.cnr.it)

Version: 3 Date: 6 March 2013

Author's response to reviews: see over
Editors of “BMC Medical Genetics”

Re: MS: 1086602316838228

Title: Sequencing of NOTCH1, GATA5, TGFBRI and TGFBR2 genes in familial cases of Bicuspid Aortic Valve

Dear Editors,

we are grateful to Reviewers for useful and constructive suggestions. In the new version we tried to resolve the last issues raised by reviewers and answer accordingly. The manuscript has been edited for grammar and wording by a native-English speaker with scientific expertise.

We hope that the revised manuscript adequately addressed all criticisms.

Yours sincerely,

Dr. Ilenia Foffa